Author's response to reviews

Title: Successful outcome of a pregnancy in advanced cirrhosis due to Hepatitis B surface antigenemia, delta super-infection and Hepatitis C co-infection: a case report

Authors:

Dr AMNA Subhan (amna.subhan@aku.edu)
Dr SHAHAB Abid (shahab.abid@aku.edu)
Dr WASIM Jafri (wasim.jafri@aku.edu)

Version: 3 Date: 16 August 2007

Author's response to reviews: see over
Dated; Aug 8, 2007

Editorial Office,
Journal of Medical case reports,
Dear Editor,

Thanks to review our manuscript entitled “Successful outcome of a pregnancy in advanced cirrhosis due to Hepatitis B surface antigenemia, delta super-infection and Hepatitis C co-infection: a case report”. We really appreciate the reviewer’s comments. All comments were read carefully and changes done in the revised manuscript according to suggestions of the reviewers. Following are the response to the concerns raised by reviewers:

Reviewer 1:

1. Ultrasound abdomen and endoscopic images of esophageal varices are submitted with the revised manuscript.
   
2. Diagnosis of Cirrhosis was based upon presence of clinical stigmata for chronic liver disease and supported by hematological, biochemical parameters, sonographic and endoscopic findings. Liver biopsy was not done in this case.

3. Yes, the labour was normal in this case.

4. Despite thrombocytopenia and coagulopathy there was no episode of gastrointestinal or postpartum bleed.

5. References are checked again and changed according to the style recommended.


Reviewer 2:

1. Manuscript revised

2. Language and spelling errors corrected.
We are once again thankful to the whole team of JMCR and the reviewers. Their valuable suggestions gave a better shape to our case report and elaborated further the message, which we want to communicate to the readers of JMCR. We hope that the revised manuscript will now be acceptable for publication.

Best regards,
Yours Sincerely
Dr Amna Subhan
MBBS, FCPS
Fellow Gastroenterology,
Aga Khan University Hospital,
Stadium Road, Karachi,
Pakistan.
Tel. #: 92-21-486-4659
Fax #: 92-21-493-4294
e-mail: amna.subhan@aku.edu